Author's response to reviews

Title: Antitumor effectiveness of different amounts of electrical charge in Ehrlich and fibrosarcoma Sa-37 tumors

Authors:

Hector C Ciria (camue@cnea.uo.edu.cu)
Magaly S Quevedo (camue@cnea.uo.edu.cu)
Luis B Cabrales (berguesc@yahoo.com)
Rodolfo P Bruzon (naogit@yahoo.com)
Miriam F Salas (camue@cnea.uo.edu.cu)
Odalis G Pena (amandaernesto@yahoo.es)
Tamara R Gonzalez (camue@cnea.uo.edu.cu)
Demetrio S Lopez (camue@cnea.uo.edu.cu)
Jose M Flores (jmarin@cajeme.cifus.uson.mx)

Version: Date: 1 November 2004

Author's response to reviews:

In Background, P2, line #8, we insert "increased" before "survival rate". Also, in P3, line #10, "clinic" was changed to "clinical"

In Discussion, P2, line #9, the sentence "On the other hand, Von Euler [11] observed apoptosis around the anode and both necrosis and apoptosis around cathode" was changed by the sentence "On the other hand, Von Euler [11] observed both apoptosis and necrosis around the anode but only necrosis around cathode"

References:
In reference 1, BEW was changed by BE
in reference 2, the word "and" was removed from the author list
in reference 8, NJF was changed by NJ
in reference 9, the word "and" was removed from the author list
in reference 12, NJF was changed by NJ
In table 2, "monocites infiltrate" was changed by "monocytic infiltrate"
table's title and legend were placed above the table that it represents